Author's response to reviews

Title: Physicians' propensity to collaborate and their attitude towards EBM: A cross-sectional study

Authors:

Daniele Mascia (dmascia@rm.unicatt.it)
Americo Cicchetti (acicchetti@rm.unicatt.it)
Maria Pia Fantini (mariapia.fantini@unibo.it)
Gianfranco Damiani (gdamiani@rm.unicatt.it)
Walter Ricciardi (wricciardi@rm.unicatt.it)

Version: 6 Date: 12 July 2011

Author's response to reviews:

July, 12th 2011

Dear Editorial Production Team,

thank you for the invitation to revise and resubmit our paper titled “PHYSICIANS’ PROPENSITY TO COLLABORATE AND THEIR ATTITUDE TOWARDS EBM: A CROSS-SECTIONAL STUDY”. We modified our paper according to the editorial requests. Detailed point-to-point responses follow. All the requested documents are been accompanied electronically.

Yours sincerely,
Daniele Mascia

Response to the Editorial production team

-----------------------------------------------

Major revisions (we require the author to make these changes)

Footnotes: We are unable to accommodate footnotes (such as that on page 6) in the full text version of the manuscript. We suggest that you incorporate the footnotes into the text of the manuscript. Alternatively, you may wish to make an appendix section containing your footnotes

** As suggested, we incorporated footnotes in the text of the paper.

Additional files: all additional files must be mentioned in the text in numerical order, or removed from the system. Please ensure all files are referred to as additional files (not, for example, supplementary data)

** We removed the additional files from the system.

References: References should be cited in the text using consecutive numbers in square brackets. Currently, reference 31 follows reference 29. Reference 35 also
seems to be uncited. Also, please place the reference before the full stop. For example 'text [1].'

** We double-checked all references. In the revised manuscript, reference 30 follows reference 29 and reference 35 is cited.

----------------------------
Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Structure: Please change the heading "Research methods" to "Methods".

** Text changed accordingly.

Highlighting: Please remove all highlighting form the manuscript.

** Highlighting removed.

Tables: Please remove all vertical lines from tables.

** Vertical lines removed.